Author's response to reviews

Title: Central poststroke pain: somatossensory abnormalities and the presence of associated myofascial pain syndrome.

Authors:

Rogério A.A. de Oliveira (rogerio.adas@uol.com.br)
Daniel Ciampi de Andrade (ciampi@usp.br)
André G.G. Mchado (aggmachado@yahoo.com)
Manoel J. Teixeira (manoelj@aclnet.com.br)

Version: 4 Date: 11 August 2011

Author's response to reviews: see over
São Paulo, August 5th, 2011

To the Editors of BMC Neurology

Please find enclosed to this letter our manuscript on the clinical evaluation of Central Post-Stroke Pain patients and the role of miofascial pain syndrome.

Stroke ranks among the leading causes of death in the world, and up to 40% of these patients will experience Post-Stroke Pain. Among the causes of PSP is CPSP, a neuropathic pain syndrome present in 11% of stroke patients. The definition of CPSP is currently being debated, and some proposals to change it have been made recently. However, main points remain unknown. Very little is known on the co-occurrence of CPSP and other PSP syndromes, and to which extent this association is related to more severe pain or functional impairment in these patients.

We assessed pain characteristics and somatosensory deficits with standardized tools in 40 CPSP and evaluated the presence of other PSP syndromes, in particular, myofascial pain syndrome (MPS). We found that the presence of MPS is frequent in CPSP patients, although its presence was not related to more severe pain in this population. Our results strongly suggest CPSP is more frequently associated with other PSP than and “isolated” entity, and that future diagnostic criteria should take this information into account.

We hope these findings and others enclosed in the paper might merit publication in BMC Neurology should it meet the standards of your Journal.

All authors have read the manuscript, which has never been published and is not under simultaneous evaluation by another journal.

Sincerely;

Dr. Daniel Ciampi de Andrade

Divisão de Clínica Neurológica do Hospital das Clínicas da FMUSP
Secretaria da Neurologia, Instituto Central
Av.Dr. Enéas de Carvalho Aguiar, 255, 5º andar, sala 5084 – Cerqueira César
05403-900 – São Paulo – SP – Brazil . + 5511 9775 3538 ciampi@usp.br